MOSS 1 Debrief Minutes

May 13, 2003

ELEMENT ATTENDANCE:

ESDIS SSDO, ESDIS SOO, EDOS, EMOS, ASDC DAAC, TES SIPS, GES DAAC, MLS SIPS, HIRDLS SIPS, OMI SIPS, Averstar

NEXT TELECON: May 27, 2003 at 2pm EST in Bldg 32 room S241

DIAL-IN NUMBER:

USA Toll Free Number: 888-395-8106

PASSCODE: 92854

MOSS 2 (3 day test) Data Preparation Schedule:
–May 19-June 6: MOSS 2 data delivery and preparation (Science data still needed for TES, HIRDLS, and OMI)

–June 9-13: Write tapes for Polar Ground Stations

–June 16-17: Ship tapes to Polar Ground Stations

–June 30: MOSS 2 TRR

–July 1-2: (TBD) Abbreviated Dry run with Ground Stations

–July 14-18: Formal MOSS 2 test
MEETING HIGHLIGHTS:

All debrief presentations can be obtained from the Aura Testing website: http://jupiter02.gsfc.nasa.gov/AuraTesting/ under MOSS1 link. Only highlights, changes, or clarifications will be presented below.

SSDO:

Karen presented the Mission Critical Science and Overall Mission Science requirements test status as of MOSS1.  In preparation for MOSS 2, Karen asked all elements to review the requirements allocated to them and report any discrepancies or changes and to identify any work and/or testing required (ex: OMI L2 ESDTs that are not delivered yet). In addition, Karen reviewed the schedule presented above. TES stated that they have completed the first step in their data generation process and expect to deliver data to ESDIS on 5/18/03. Karen indicated that ESDIS could set the quick look flags for EDSs. TES is requesting the flags to be set – 1min per orbit per APID. HIRDLS reported that they would get back to Karen on their data delivery date, but intend to deliver more data and more APIDs. The data will have to be time-shifted. OMI mentioned that they have sent Nico of Dutch Space, Netherlands the OMI time-shifted data and will have to wait for a response from Nico about the data. In the meantime, OMI received the official L1B software and has run successfully with the test data supplied with the software.

Karen mentioned that the Aura project is requesting a science debrief for CPT1 and SCIF3 for May 21 and wanted to know if the elements could support that date or would need more time to analyze their results. Karen also asked the SIPS to try to process the data to the extent possible using ORBSIM and report any results.

Karen reported that the GES DAAC will be in OPS mode for MOSS2 and would have to install all the Aura ESDTs in that mode. Before this can happen, the ESDTs first need to be validated and certified (required by CCB).  For validation of the ESDTs, only the original .met file representing a granule for each ESDT is needed. James Johnson sent out an email, after the telecon, informing the SIPS of where they could ftp their data. This data should also include any new products that have not yet been ingested at the DAAC.

EDOS:

Ron reported that their secure ftp implementation is scheduled for internal testing around mid June and external testing will occur afterwards, around mid-late June. He mentioned that the testing strategy would be to test with HIRDLS NRT first and the other systems would be based on customer request.

EMOS:

Utah presented his slides via telecon. No issues/problems were found during processing or distributing files to the GES DAAC. EMOS received all the data that were expected.

ASDC DAAC:

Sue presented the slides via telecon. Sue reported that ASDC/OPS mode ingested all TES L0 granules from EDOS and DPREP products from the GES DAAC. The issues/problems found were TES L1 Ingest failures but were later resolved (2 metadata problems) and the products ingested without problems. Karen recommended that since Averstar was able to search and order the TES L1 products via EDG, then requirement #10 (distribution of TES L1 data products) should pass. Steve indicated that he would update the database with a comment and pass the requirement.

TES SIPS:

Doug presented the slides via telecon for Andy Griffin. The issues/problem found were related to the distribution of the L1 products. L1 production was not planned for MOSS1. TES SIPS successfully distributed L2 products to the DAAC, therefore receiving bonus points for MOSS1.

GES DAAC:

Ed reported that the GES DAAC experienced no problems during MOSS1. The GES DAAC received and distributed all products expected. Ed mentioned that TS2 mode would soon be unavailable due to install of 6A.07 in that mode.

MLS SIPS:

Audrey presented the slides via telecon. MLS reported that the ML1OA product did not have meaningful content; the software bypassed the processing of the MOSS1 DPREP HDF output due to the toolkit problems. MLS created and distributed all of their L1 products.

HIRDLS:

Vince presented his slides via telecon. No issues/problems reported for MOSS1. Rate buffer delivery to HIRDLS NRTP from EDOS was not planned for MOSS1; awaiting EDOS secure ftp implementation. Karen acknowledged that HIRDLS used the binary version of the MOSS1 DPREP output for their processing.

OMI SIPS:

Marghi presented the slides via telecon. No issues/problems reported for MOSS1. The requirement to produced L1 products from L0 products will be tested in MOSS2. However, OMI received bonus points for distribution of all their L1Bs and two L2s products to the GES DAAC.

Averstar:

Dot presented the slides for Scott Applebee via telecon. Averstar was able to, via EDG, search and order HIRDLS, MLS, OMI, and TES products. Two problems were encountered (1) GES DAAC/TS2 -- User password appeared in V0GWY debug log (NCR ESCed37450) and (2) ASDC DAAC/OPS -- Order placed for an unsupported media type (D3 tape). Glenn Iona asked Averstar to check to see if this problem (#2 above) exists at the other DAACs. After the telecon, Sue at ASDC DAAC wrote trouble ticket LRC000000005954 to ensure permanent solution and Latha at GES DAAC is going to verify that D3 is no longer an option and modify the configuration file accordingly. 

